Title: Primary mucinous adenocarcinoma of the thymic gland with intestinal differentiation and HLA-DRB5-locus deletion. An immunohistochemical and molecular study with review of the Literature

Authors:

Maryam Maghbool (maghbol.maryam@yahoo.com)
Mani Ramzi (ramzim@sums.ac.ir)
Inga Nagel (inagel@medgen.uni-kiel.de)
Pablo Bejarano (pbejaran@med.miami.edu)
Reiner Siebert (rsiebert@medgen.uni-kiel.de)
Yahya Daneshbod (y@daneshbod.com)
Abolfazl Saeedzadeh (asaeedzadeh@gmail.com)

Version: 4 Date: 19 May 2013

Author's response to reviews:

Dear Chief Editor
Thank you very much for accepting our paper and reviewers point of view for improving our paper.
The following changes are made according to reviewer 2.
Table 1 is added.